\documentclass[11pt,reqno]{amsart}
\usepackage[cp1251]{inputenc}
\usepackage[english,russian]{babel}
\usepackage{amsmath}
\usepackage{amssymb}
\usepackage{amsfonts}
\usepackage{graphicx}
\usepackage{xcolor}
\usepackage[colorlinks]{hyperref}

\usepackage{titlesec}
\titlespacing{\section}{0pt}{2.5ex}{1.5ex}
\titlespacing{\subsection}{0pt}{1.5ex}{1ex}
\titlespacing{\subsubsection}{0pt}{1.5ex}{1ex}
\titleformat{\section}{\large\bfseries\centering}{\thesection}{1em}{}
\titleformat{\subsection}[runin]{\bfseries}{\thesubsection.}{0.5em}{}[.\mbox{\ }]
\titleformat{\subsubsection}[runin]{\bfseries}{\thesubsubsection.}{0.4em}{}[.\mbox{\ }]


%%%%%---------- ORCiD
\newcommand\orcidicon[1]{\href{https://orcid.org/#1}{\includegraphics[scale=0.02]{orcid.pdf}}}


\def\udcs{517.957, 517.929} %Here the author places classificators of the paper according to Russian classification system
\def\mscs{35C05, 35K55, 35K57, 35R10} %Here the author places  classificators according to the AMS classification list.
\setcounter{page}{144}

\newtheorem{lemma}{Lemma}
\newtheorem{theorem}{Theorem}
\newtheorem{definition}{Definition}
\newtheorem{corollary}{Corollary}
\newtheorem{proposition}{Proposition}
\newtheorem{state}{State}


\hypersetup{linkcolor=blue,filecolor=black,urlcolor=blue, citecolor=blue}

\def\logo{\textcolor{blue}{{\bf\Huge S\raisebox{0.2ex}{\hspace{0.55ex}\raisebox{0.05ex}e\hspace{-1.65ex}$\bigcirc$}MR}}}

\def\semrtop
     {
  \vbox{
     \noindent\logo \hspace{-3mm}  \parbox{12cm}{     \begin{center}
     {\LARGE \textsf{СИБИРСКИЕ ЭЛЕКТРОННЫЕ}} \\[2mm]
     {\LARGE \textsf{МАТЕМАТИЧЕСКИЕ ИЗВЕСТИЯ}} \\[2mm]
     {\Large \textsf{Siberian Electronic Mathematical Reports}} \\[1mm]
     {\Large\tt{\textcolor{blue}{http://semr.math.nsc.ru}}}\\[1mm]
     \raisebox{1ex}{ISSN 1813-3304 }
     \vspace*{-3mm}
%     {\small Институт математики им. С.Л.Соболева СО РАН}\\[-1mm]
%     {\small Sobolev Institute of Mathematics SB RAS}
     \end{center}
}
     \vspace{-1mm}
     \noindent
      \begin{tabular}{c}
     \hphantom{aaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaa!} \\
     \hline\hline
     \end{tabular}

     \vspace{3mm}
     {\flushleft\it Vol. 20, No. 2, pp. 144--145 (2023) \hspace{\fill}{\rm\small УДК \udcs \hspace{-4mm}}}
     \newline  %Volume, pages in Russian, filled by Editorial board
     	{\rm\small \textcolor{blue}{https://doi.org/10.33048/semi.2023.20.???}}\hphantom{aaaaaaaaaaaaaaaaaaaaaaaa\;}{\rm\small MSC\  \ \mscs }
  }%\vbox
}

\begin{document}
\begin{thebibliography}{99} % Ссылки на литературу приводятся только на английском языке (даже если у источника нет англоязычной версии).

\bibitem{atm} Aibinu M.O., Thakur S.C., Moyo S. {\it On Construction of Exact Solutions of Delay Reaction-Diffusion Systems}, Proceedings of the Conference Analytical and Numerical Methods in Differential Equations, (ANMDE 2021 and Yanenko 100) Suranaree University of Technology, Thailand August 23--26, 2021.

\bibitem{aa2021}  Aleksandrov A. {\it A problem of formation control on a line segment under protocols with communication delay}, Systems $\&$ Control Letters, {\bf 155} (2021), 104990.

\bibitem{ap2021} Aksenov A.V., Polyanin A.D. {\it Methods for constructing complex solutions of nonlinear PDEs using simpler solutions}, Mathematics,  \textbf{9}:345 (2021), 1--30. https://doi:10.3390/math9040345.

\bibitem{vj1996} Corless R.M, Gonnet G.H., Hare D.E., Jeffrey D.J. and Knuth D.E. {\it On the Lambert $W$ function}, Advances in Computational Mathematics, {\bf 5} (1996), 329--359.

\bibitem{eb} Elamvazhuthi K., Berman S. {\it Mean-field models in swarm robotics: a survey}, Bioinspiration and
Biomimetics,  \textbf{15}:1 (2019), 015001. https://doi.org/10.1088/1748-3190/ab49a4

\bibitem{gant1988} Gantmaher F.R. {\it Matrix theory}, AMS Chelsea Publishing: An Imprint of the American Mathematical Society, 1959; 660 p.

\bibitem{gor1988} Goryachenko V. D. {\it Qualitative Methods in the Dynamics of Nuclear Reactors}, M.: Energoatomizdat, 1983. (In Russian)

\bibitem{gkf2010} Gradetskiy V.G., Knyaz'kov M.M., Fomin L.F., Chashchukhin V.G. {\it Miniature robot mechanics},  M.: Nauka, 2010. (In Russian)

\bibitem{hsy} Hsu C.-H., Yang T.-S., Yu Z.-X. {\it Existence and exponential stability of traveling waves for delayed reaction-diffusion systems}, Nonlinearity, 2018. Vol. 31, pp. 838--863.

\bibitem{kb2008} Kirillov P.L., Bogoslovskaya G.P.  {\it Heat and mass transfer in nuclear power plants. 2nd ed.}, M.: IzdAT, 2008, 256 p. (In Russian)

\bibitem{ks2017smz}  Kosov  A.A., Semenov E.I. {\it Multidimensional exact solutions to the reaction-diffusion system with power-law nonlinear terms}, Siberian Math. J.,  \textbf{58}:4  (2017), 619--632.  https://doi: 10.17377/smzh.2017.58.408

\bibitem{ks2018du} Kosov  A.A., Semenov E.I. {\it On Exact Multidimensional Solutions of a Nonlinear System of Reaction–Diffusion Equations}, Differential Equations,  \textbf{54}:1 (2018), 106--120.  https://doi: 10.1134/S0374064118010090

\bibitem{ks2021} Kosov A.A., Semenov E.I. {Distributed model of space exploration by two types of interacting robots and its exact solutions}, Journal of Physics. Conference Series, 2021. Vol. 1847(1), 012007.   https://doi:10.1088/1742-6596/1847/1/012007

\bibitem{lyx} Liu Y., Yu Zh., Xia J. {\it Exponential stability of traveling waves for non-monotone delayed reaction diffusion equations}, Electronic Journal of Differential Equations, {\bf 86} (2016), 1--15. URL: http://ejde.math.txstate.edu

\bibitem{lw2010} Lv G.Y., Wang M. {\it Nonlinear stability of travelling wave fronts for delayed reaction diffusion equations}, Nonlinearity, {\bf 23} (2010), 845--873.

\bibitem{moz} Mei M., Ou C., Zhao X.-Q. {\it Global stability of monostable traveling waves for nonlocal time-delayed reaction-diffusion equations}, SIAM J. Math. Anal., {\bf 42} (2010), 2762--2790.

\bibitem{mm2008} Meleshko, S.V., Moyo, S. {\it On the complete group classification of the reaction-diffusion equation with a delay}, J. Math. Anal. Appl., {\bf 338} (2008), 448--466.

\bibitem{mur} Murray J.D. {\it Mathematical biology. I. An Introduction}, Springer, 2002, 552 p.

\bibitem{pll} Pan S, Li W.-T., Lin G. {\it Existence and stability of traveling wave fronts in a nonlocal diffusion equation with delay}, Nonlinear Anal., {\bf 72} (2010), 3150--3158.

\bibitem{pz2014} Polyanin A.D., Zhurov A.I. {\it Exact separable solutions of delay reaction-diffusion equations and other nonlinear partial functionaldifferential equations}, Commun. Nonlinear Sci. Numer. Simul., {\bf 19} (2014), 409--416.

\bibitem{pz2014a} Polyanin A.D., Zhurov  A.I. {\it New generalized and functional separable solutions to nonlinear delay reaction-diffusion equations}, Int. J. Non-Linear Mech., 2014. Vol. 59. pp. 16--22.

\bibitem{ps2015} Polyanin A.D., Sorokin V.G {\it Nonlinear delay reaction-diffusion equations: Traveling-wave solutions in elementary functions}, Appl. Math. Lett., {\bf 46} (2015), 38--43.

\bibitem{p2019} Polyanin, A.D. {\it Generalized traveling-wave solutions of nonlinear reaction-diffusion equations with delay and variable coefficients}, Appl. Math. Lett., {\bf 90} (2019), 49--53.

\bibitem{ps2021a} Polyanin, A.D., Sorokin V.G. {\it A method for constructing exact solutions of nonlinear delay PDEs}, J. Math. Anal. Appl., {\bf 494} (2021). 

\bibitem{ps2021} Polyanin A.D., Sorokin V.G. {\it Nonlinear pantograph-type diffusion PDEs: exact solutions and the principle of analogy}, Mathematics, \textbf{9}:51 (2021), 1--23. https://doi: 10.3390/math9050511

\bibitem{pz2020} Polyanin A.D., Zhurov A.I. {\it Separation of variables and exact solutions to nonlinear PDEs}, CRC Press. Taylor $\&$ Francis Group. 2022.

\bibitem{psz2022} Polyanin A.D., Sorokin V.G., Zhurov A.I. {\it Delay ordinary and partial differential equations}, CRC Press. Taylor $\&$ Francis Group. 2023.

\bibitem{ss2020}  Su Si, Zhang Guo-Bao. {\it Global stability of traveling waves for delay reaction-diffusion systems without quasi-monotonicity}, Electronic Journal of Differential Equations, {\bf 46} (2020), 1--18.

\bibitem{sv2022} Sorokin V.G., Vyazmin A.V. {\it Nonlinear Reaction-Diffusion Equations with Delay: Partial Survey, Exact Solutions, Test Problems, and Numerical Integration}, Mathematics, \textbf{10}:1886 (2022). https://doi.org/10.3390/math10111886

\bibitem{dds2006} Valluri S.R., Jeffrey D.J., Corless R.M. {\it Some applications of the Lambert $W$ function
to physics}, Canadian J. Physics, {\bf 78} (2000), 823--831.

\bibitem{wfj}  Wei J., Fridman E., Johansson K.H. {\it  A PDE approach to deployment of mobile agents under leader relative position measurements}, Automatica, {\bf 106} (2019), 47--53.

\bibitem{wu2011} Wu S.-L., Zhao H.-Q., Liu S.-Y. {\it Asymptotic stability of traveling waves for delayed reaction-diffusion equations with crossing-monostability}, Z. Angew. Math. Phys., {\bf 62} (2011), 377--397.

\end{thebibliography}

\renewcommand{\refname}{References}
\renewcommand{\proofname}{Proof.}
\renewcommand{\figurename}{Fig.}

\thispagestyle{empty}

\title[On exact solutions of delay PDEs]
{\large On exact solutions of delay PDE{\small s} used in modeling of distributed formations}

\author[A.A. KOSOV]{{\bf A.A. Kosov}\protect\orcidicon{0000-0003-1352-1828}}
\author[E.I. SEMENOV]{{\bf E.I. Semenov}\protect\orcidicon{0000-0002-9768-9945}}

\address{Alexander Arkad'evich. Kosov  % Контактные данные всех авторов и место работы указываются только на английском языке
\newline\hphantom{iii} Matrosov Institute for System Dynamics and Control Theory, Siberian Branch of Russian Academy of Sciences, ISDCT SB RAS,
\newline\hphantom{iii} Lermontov str., 134,
\newline\hphantom{iii} 664033, Irkutsk, Russia}%
\email{\textcolor{blue}{kosov\_idstu@mail.ru}}%

\address{Eduard Ivanovich. Semenov  % Контактные данные всех авторов и место работы указываются только на английском языке
\newline\hphantom{iii} Matrosov Institute for System Dynamics and Control Theory, Siberian Branch of Russian Academy of Sciences, ISDCT SB RAS,
\newline\hphantom{iii} Lermontov str., 134,
\newline\hphantom{iii} 664033, Irkutsk, Russia}%
\email{\textcolor{blue}{edwseiz@gmail.com}}%

\thanks{\sc Kosov, A.A., Semenov E.I., On exact solutions of delay PDEs used in modeling of distributed formations}
\thanks{\copyright \ 2023 Kosov A.A., Semenov E.I}
\thanks{\rm This work was carried out with financial support from the Ministry of Science and Higher Education of the Russian Federation (projects № 121041300058-1 and № 117032210078-4)}
\thanks{\it Received January, 1, 2023, Published December, 31, 2023}%

\semrtop \vspace{1cm}
\maketitle {\small

\vspace{-12pt}
\centerline{{\it Communicated by} {\sc A.A. Kosov}
%({\it filled by Editorial board})
}

\bigskip
\bigskip

\begin{quote}
\noindent{\bf Abstract:} 
In this paper we study a multidimensional nonlinear differential equation of parabolic type with delay, considered as a mathematical model for changing the density of a distributed formation of autonomous robots. The problem of constructing exact solutions is considered. The variant of reduction method, allowing to separate search of dependence of the solution on time and on spatial variables, is offered. The dependence of solutions on spatial variables is found from a system of algebraic equations, and the time dependence is found from a system of ordinary differential equations with delay. A number of examples are given for the construction of exact solutions, which are explicitly expressed in terms of elementary and Lambert functions.
\medskip

\noindent{\bf Keywords:} equations of parabolic type; delay PDEs; exact solutions; formation modeling; Lambert function.
 \end{quote}
}

\bigskip

\section{Introduction and problem statement}

One of the main trends in the development of modern robotics and multi-agent systems is the miniaturization of autonomous vehicles, combining them into <<warms>> and <<flocks>> for interaction in order to increase the efficiency and success of solving problems and completing missions~\cite{gkf2010}. This trend inevitably leads to the need to use distributed characteristics in mathematical modeling, such as densities, concentrations, etc. and application of partial differential equations. Everything here is similar to how macroparameters (temperature, pressure, etc.) are successfully used for description in fluid and gas mechanics. Partial differential equations began to be used in modeling the movement of formations of moving objects with distributed characteristics. Thus, in~\cite{wfj}, to model a formation with control based on the principle of deviation from the leader's movement, a parabolic type equation of the following form was used
\begin{equation}\label{Eq1}
u_t=F\left(u,u_{x},u_{xx}\right),\quad t\geq 0,\quad x\in\mathbb{R}. 
\end{equation}
In~\cite{ks2021}, a distributed model of three-dimensional space exploration described by a system of two parabolic-type equations was proposed by two types of robots that propagate into the surrounding space from a certain common base and interact with each other. In some cases, a certain time is spent on the exchange of information between robots and the leader and the formation of a feedback signal, and it becomes necessary to take into account the delay in the equations of motion. For example, in~\cite{aa2021}, taking into account the delay, the problem of equidistant distribution of autonomous robots on a given segment of a straight line was considered. Problems of this kind with the need to take into account the delay in the general case lead to the study of equations of the form
\begin{equation}\label{Eq2}
u_t(t,x)=F\left(u(t-\tau,x),u(t,x),u_{x}(t,x),u_{xx}(t,x)\right),
\end{equation}
\begin{equation*}
\quad t\geq 0,\quad\tau>0,\quad x\in\mathbb{R}.
\end{equation*}
Note that many models and problems that arise in biology, chemistry, economics, and other fields also lead to equations of the form (\ref{Eq2}), where the rate of a process may depend not only on the current, but also on previous states. The presence of delay in the nonlinear equation (\ref{Eq2}) significantly complicates the study and construction of exact solutions. A number of results in this direction were obtained in~\cite{ap2021,mm2008,pz2014,pz2014a,ps2015,p2019,ps2021a,ps2021}, the most studied are equations with one space variable. A detailed summary of the results and an extensive bibliographic list of more than 600 papers on the construction of exact solutions for partial differential equations with delay are given in the monograph~\cite{psz2022}. We also note a detailed (219 sources cited) survey~\cite{sv2022} on exact solutions and methods of numerical integration of a delayed PDEs. The existence of traveling waves and various types of their stability (asymptotic, exponential, global) for reaction-diffusion systems with delay were studied in papers~\cite{atm,hsy,lyx,lw2010,moz,pll,ss2020,wu2011}.

In this article, we will consider a multidimensional special case of the equation (\ref{Eq2}) of the following form
\begin{equation}\label{1}
u_{t}=\nabla\cdot\biggl(u^{\lambda}\nabla u\biggr)+\alpha\bar{u},
\end{equation}
where $u=u({\bf x},t)$, $\bar{u}=u({\bf x},\bar{t})$, $\bar{t}=t-\tau$, ${\bf x}\in\mathbb{R}^n$; $\tau>0$; $\nabla$~--- gradient operator relative to ${\bf x}\in\mathbb{R}^n$; $\lambda\ne 0$~--- real parameter of medium nonlinearity; $\alpha\ne 0$~--- some constant. In the equation (\ref{1}), $u({\bf x},t)$ represents the formation density at time $t$ at a point in space ${\bf x}\in\mathbb{R}^n$, therefore, only non-negative solutions $u({\bf x},t)\geq 0$ are of interest. The left side in (\ref{1}) is the rate of formation density change over time, the first term on the right side describes the nonlinear diffusion process of formation propagation in space, and the second term is a distributed control action formed with some informational delay $\tau >0$ according to the principle of linear feedback. Wherein, in the case of negative feedback $\alpha<0$, when zero density is reached, the control signal is switched off, i.e. $\bar{u}=0$ for $u({\bf x},t)=0$.

Note that the above interpretation of the equation (\ref{1}) is certainly not the only one. This equation can also be considered as an equation of nonlinear heat conduction with a linear source (sink), in which the process of heat release (or absorption) occurs with some delay $\tau>0$. An example of such a process is heat transfer in a nuclear reactor~\cite{gor1988,kb2008}. The equation (\ref{1}) can also be considered as a mathematical model of the distribution of a biological species over the habitat~\cite{mur}, while the second term on the right side (\ref{1}) takes into account reproduction, and the value $\tau>0$ represents breeding time. Other interpretations of the equation (\ref{1}) are also possible.

Thus, for many applications it is of interest to study the equation (\ref{1}), construct its exact solutions, and study their properties. The main purpose of this work is to construct exact multidimensional solutions of the equation (\ref{1}) in the form of a generalized separation of variables. To do this, we transform (\ref{1}) by a simple replacement $u^{\lambda}=v$ to the equation for the new desired function $v$
\begin{equation}\label{2}
v_{t}=v\Delta v+\nu|\nabla v|^2+\alpha\lambda v^{1-\nu}\bar{v}^{\nu},
\end{equation}
where $\nu=1/\lambda$, $v=v({\bf x},t)$, $\bar{v}=v({\bf x},\bar{t})$, $\Delta$~--- the Laplace operator in $\mathbb{R}^n$. 
Exact multidimensional solutions of the equation (\ref{2}) will be sought in the form of a multiplicative separation of variables~\cite{pz2020}
\begin{equation}\label{3}
v({\bf x},t)=\psi(t)W({\bf x}), 
\end{equation}
with multidimensional function $W({\bf x})$ the following kind
\begin{equation}\label{4}
W({\bf x})=\frac{1}{2}(A{\bf x},{\bf x})+({\bf B},{\bf x})+C, 
\end{equation}
where is the function $\psi(t)\in C^{1}([0,+\infty))$, non-zero numeric symmetric matrix $A$ of size $n\times n$, the constant vector ${\bf B}\in\mathbb{R}^n$ and the constant $C\in\mathbb{R}$ are to be determined. In addition, in the case of $\lambda=1$, multidimensional solutions of the equation (\ref{1}) will be sought in the form of an additive separation of variables~\cite{pz2020}
\begin{equation}\label{5}
u({\bf x},t)=\varphi(t)+W({\bf x}), 
\end{equation}
where the function $\varphi(t)\in C^{1}([0,+\infty))$ and the function $W({\bf x})$ has the form (\ref{4}).

\section{Main results}

After substituting the function (\ref{3}) into the equation (\ref{2}) and elementary transformations, we arrive at the expression
\begin{equation*}
\psi'W=\psi^2W\Delta W+\frac{1}{\lambda}\psi^2|\nabla W|^2+\alpha\lambda\psi^{1-\frac{1}{\lambda}}\bar{\psi}^{\frac{1}{\lambda}}W.
\end{equation*}
Here and below $\psi'=\displaystyle\frac{d\psi(t)}{dt}$, $\bar{\psi}=\psi(\bar{t})\equiv\psi(t-\tau)$. Since $W({\bf x})$ is defined by the formula (\ref{4}), by direct calculation we find
\begin{equation*}
|\nabla W({\bf x})|^2=(A^2{\bf x},{\bf x})+2(A{\bf B},{\bf x})+|{\bf B}|^2, 
\end{equation*}
\begin{equation}\label{6}
\Delta W({\bf x})={\rm tr}\,A~- \mbox{matrix trace}\;A. 
\end{equation}
Taking into account the relations (\ref{6}), we rewrite the last equality as
\begin{equation*}
\left(\psi'-{\rm tr}\,A \psi^2-\alpha\lambda\psi^{1-\frac{1}{\lambda}}\bar{\psi}^{\frac{1}{\lambda}}\right)\left(\frac{1}{2}(A{\bf x},{\bf x})+({\bf B},{\bf x})+C\right)=
\end{equation*}
\begin{equation}\label{7}
=\frac{1}{\lambda}\psi^2\left((A^2{\bf x},{\bf x})+2(A{\bf B},{\bf x})+|{\bf B}|^2\right).
\end{equation}
\begin{lemma}
If the numerical symmetric matrix $A$, the vector ${\bf B}\in\mathbb{R}^n$ and the constant $C\in\mathbb{R}$
satisfy the system of algebraic equations $(${\it SAE}$)$
\begin{equation}\label{8}
A=2\sigma\,A^2, \quad
{\bf B}=2\sigma\,A{\bf B}, \quad
C=\sigma\,|{\bf B}|^2, 
\end{equation}
where $\sigma\ne 0$~--- separation constant, then the equality $(\ref{7})$ reduces to an ordinary differential equation $({\it ODE})$ with a delay of the following form:
\begin{equation}\label{9}
\psi'(t)=\left({\rm tr}\,A+\frac{1}{\lambda\sigma}\right)\psi^2(t)+\alpha\lambda\psi^{1-\frac{1}{\lambda}}(t)\psi^{\frac{1}{\lambda}}(t-\tau).
\end{equation}
\end{lemma}
\begin{proof}
If the numeric symmetric matrix $A$, the vector ${\bf B}\in\mathbb{R}^n$ and the constant $C\in\mathbb{R}$ satisfy SAE (\ref{8}), then (\ref{7}) can be rewritten as
\begin{equation*}
\left[\psi'(t)-\left({\rm tr}\,A+\frac{1}{\lambda\sigma}\right)\psi^2(t)-\alpha\lambda\psi^{1-\frac{1}{\lambda}}(t)\psi^{\frac{1}{\lambda}}(t-\tau)\right]\times
\end{equation*}
\begin{equation*}
\times\biggl[(A^2{\bf x},{\bf x})+2(A{\bf B},{\bf x})+|{\bf B}|^2\biggr]=0.
\end{equation*}
This equality turns into an identity when the expression in the first square bracket, which is the ODE (\ref{9}), vanishes. The lemma is proven.
\end{proof}

Thus, the construction of exact multidimensional solutions of the nonlinear heat equation with a linear delayed source (sink) (\ref{1}) using the ansatz (\ref{3}) has been reduced to integrating an ODE with delay (\ref{9}) and the solvability of the system algebraic equations SAE (\ref{8}).

We note that the SAE (\ref{8}) was previously studied by the authors in~\cite{ks2017smz,ks2018du} in the study and construction of exact multidimensional solutions of the nonlinear system of reaction-diffusion equations using the construction (\ref{4}).
Therefore, we will not repeat its complete study, but only present the solution of this system. The matrix
\begin{equation}\label{10}
A=\displaystyle\frac{1}{2\sigma}\,SE_{m}S^{T},
\end{equation}
is a solution to the matrix equation (\ref{8}). Here $E_{m}$~--- is a diagonal matrix with $m\in\{1,2,\ldots ,n\}$ ones and $n-m$ zeros on the diagonal, $S$~-- is an arbitrary orthogonal matrix. At the same time, we have
\begin{equation} \label{11}
{\rm tr}\,A=\frac{m}{2\sigma},\quad m\leq n,\quad n\in\mathbb{N},\quad n\geq 2. 
\end{equation}
The vector equation (\ref{8}) is a system of $n$ linear homogeneous algebraic equations with respect to the components $b_1,\ldots ,b_n$ of the required vector ${\bf B}$. For any fixed matrix $A$ of the form (\ref{10}) with ${\rm rank}\, A=m<n$, there always exists a non-trivial solution of the linear homogeneous system, with $m$ components of the vector ${\bf B} $ can be chosen arbitrarily from $m$~--- dimensional linear manifold. In the case when ${\rm rank}\,A=m=n$, i.e. for $E_{m}\equiv E$, the linear homogeneous system of equations has a solution~--- an arbitrary vector ${\bf B}\in\mathbb{R}^{n}$. After successively finding solutions to the matrix and vector equations, the constant $C$ is uniquely determined from the scalar equation (\ref{8}).

Thus, we have arrived at the validity of the following main result of this paper.

\begin{theorem}\label{Th1}
If the matrix $A$ is given by the formula $(\ref{10})$, the vector ${\bf B}$ is determined from the linear SAE ${\bf B}=2\sigma\,A{\bf B }$, and the function $\psi(t)\in C^{1}([0,+\infty))$ satisfies ODE with delay $(\ref{9})$, then the following function
\begin{equation}\label{12}
u({\bf x},t)=\left[\psi(t)\left(\frac{1}{2}(A{\bf x},{\bf x})+({\bf B},{\bf x})+\sigma|{\bf B}|^2\right)\right]^{\frac{1}{\lambda}},
\end{equation}
is an exact multidimensional solution of the equation $(\ref{1})$.
\end{theorem}

Note that the spatial structure of solutions (\ref{12}) is determined by the rank of the matrix $A$ of the form (\ref{10}). If $\mathrm{rank}\,A=m=1$, then we have <<pseudo-multidimensional>> exact solutions, that is, solutions with a linear combination of space variables. If $1<\mathrm{rank}\,A=m<n$, then we obtain exact solutions of the equation (\ref{1}) anisotropic in space variables. Finally, if $\mathrm{rank}\,A=m=n$, then we have exact solutions that are radially symmetric in space variables. In what follows, we will consider $m>1$, $m\in\mathbb{Z}$.

Now let $\lambda=1$ and the solution is constructed as an additive separation of variables (\ref{5}). In this case, the assertion is true.

\begin{state}\label{STh2} 
Nonlinear equation with delay 
\begin{equation}\label{13}
u_t=u\Delta u+|\nabla u|^2+\alpha\bar{u},\quad\bar{u}=u({\bf x},t-\tau),
\end{equation}
has an exact solution of the form $(\ref{5})$, in which the function $W({\bf x})$ is given by the formula $(\ref{4})$, a nonzero numerical symmetric matrix $A$ of size $n\times n$, the constant vector ${\bf B}\in\mathbb{R}^n$, and the constant $C\in\mathbb{R}$ satisfy SAE
\begin{equation}\label{14}
A({\rm tr}\,A+\alpha)+2A^2=0,\quad {\bf B}({\rm tr}\,A+\alpha)+2A{\bf B}=0, 
\end{equation}
\begin{equation*}
C({\rm tr}\,A+\alpha)+|{\bf B}|^2=0,
\end{equation*}
and the function $\varphi(t)\in C^{1}([0,+\infty))$ satisfies a linear ODE with delay
\begin{equation}\label{15}
\varphi'(t)={\rm tr}\,A\,\varphi(t)+\alpha\varphi(t-\tau),
\end{equation}
where $\tau>0$~--- numeric lag parameter.
\end{state}

Note that the exact solutions of the equation (\ref{13}) in the one-dimensional case, in the form of a multiplicative and additive separation of variables, were constructed in~\cite{ap2021}.

Consider the solvability of the matrix equation (\ref{14}). Obviously, it always has a trivial solution $A=0$. Therefore, in what follows, we will consider only nontrivial solutions of this equation, assuming that ${\rm tr}\,A+\alpha\neq 0$. Otherwise, the matrix equation $A^2=0$, in the class of symmetric matrices, has a solution of the real matrix $A=0$.

\begin{state}\label{EqMatrix}
Let $E_{m}$~-- be a diagonal matrix with $m\in\{1,2,\ldots ,n\}$ ones and $n-m$ zeros placed on the diagonal arbitrarily. Then the matrix
\begin{equation}\label{16}
A=\nu SE_{m}S^{T},\quad\nu=-\displaystyle\frac{\alpha}{2+m},
\end{equation}
where $S$~--- arbitrary orthogonal matrix, is a solution to the matrix equation  $(\ref{14})$.
\end{state}
\begin{proof}
The trace of the matrix $A$ defined by the formula (\ref{16}) has the form
$\mathrm{tr}\,A=m\nu\equiv-\displaystyle\frac{\alpha m}{2+m}$. Taking this relation into account, the matrix equation (\ref{14}) will be rewritten as $\nu\,A=A^2$.
It is easy to check that the solution to this matrix equation is $A=\nu\,P$, where $P$~ is an arbitrary idempotent matrix, i.e. matrix satisfying the equality $P^2=P$.
It is known~\cite{gant1988} that any idempotent matrix $P$ can be written as
$P=ME_{m}M^{-1}$, where $M$~--- arbitrary nonsingular matrix with order $n$, $E_{m}$~--- a diagonal matrix with $m\in\{1,\ldots ,n\}$ units and $n-m$ zeros on the diagonal; $E_{m}$ is also idempotent: $E_{m}^2=E_{m}$. Since we are only interested in symmetric matrices $A$, we must also take idempotent matrices $P$ as symmetric ones, i.e. $P=SE_{m}S^{T}$, where $S$~--- is an arbitrary orthogonal matrix. From here we obtain the final form of the matrix $A$ defined by the formula (\ref{16}). The assertion has been proven.
\end{proof}

Vector equations (\ref{14}) are systems of $n$ linear homogeneous algebraic equations with respect to the components $b_1,\ldots ,b_n$ of the required vector ${\bf B}$. For any fixed matrix (\ref{16}) with $\mathrm{rank}\, A=m<n$, there always exists a nontrivial solution to the linear homogeneous system, and the components $b_1,\ldots ,b_m$ of the vector ${\bf B }$ can be chosen arbitrarily from $m$~-- dimensional linear manifold. In the case when ${\rm rank}\,A=m\equiv n$, i.e. for $E_{m}\equiv E$, the linear homogeneous system of equations has a solution~-- an arbitrary vector ${\bf B}\in\mathbb{R}^{n}$. After successively finding solutions to the matrix and vector equations, the constants $C$ are uniquely determined from the scalar equation (\ref{14}).

\section{Exact Solutions to ODEs with Delay}

First, consider a linear {\rm ODE} with delay (\ref{15}), which, taking into account the formula (\ref{16}), can be written as
\begin{equation}\label{17}
\varphi'(t)=-\frac{\alpha m}{2+m}\,\varphi(t)+\alpha\varphi(t-\tau),
\end{equation}
where $m\in\{1,2,\ldots ,n\}$. The solution of this equation will be sought by the Euler method 
\begin{equation}\label{18}
\varphi(t)=\varphi_0 e^{\mu t},
\end{equation}
where $\varphi_0\neq 0$~--- arbitrary constant. In particular, we can put $\varphi_0=\varphi(t)|_{t=0}$. Substituting (\ref{18}) into (\ref{17}) we obtain the transcendental characteristic equation for determining $\mu$
\begin{equation*}
\mu=-\frac{\alpha m}{2+m}+\alpha e^{-\mu\tau}.
\end{equation*}
The solution of this transcendental equation can be expressed in terms of the special $W$--Lambert function~\cite{vj1996,dds2006}. Thus, the solution of the equation (\ref{17}) is expressed by the following formula 
\begin{equation*}
\varphi(t)=\varphi_0\exp\left(\frac{(m+2)\,{\rm LambertW}\left(\alpha \tau\exp\left(\displaystyle\frac{\alpha\tau m}{m+2}\right)\right)-\alpha\tau m}{\tau(m+2)}\,t\right),
\end{equation*}
where $\varphi_0\neq 0$~--- arbitrary constant.

{\bf Example 1.} Let $n=4$, $m=3$, then the equation (\ref{13}) in four-dimensional coordinate space with respect to
state~\ref{STh2} has an exact, spatially anisotropic, solution
\begin{equation*}
u(x_1,x_2,x_3,x_4,t)=\varphi_0\exp\left(\frac{5\,{\rm LambertW}\left(\alpha \tau\exp\left(\displaystyle\frac{3}{5}\alpha\tau\right)\right)-3\alpha\tau}{5\tau}\,t\right)+
\end{equation*}
\begin{equation*}
+\frac{\alpha}{160}\left(-13x_1^2-7x_2^2-12x_3^2-16x_4^2+6\sqrt{3}\,x_1x_2-4\sqrt{3}\,x_1x_3-12x_2x_3\right)+
\end{equation*}
\begin{equation*}
+b_1x_1+\left(-\frac{1}{3}\sqrt{3}\,b_1+\frac{2}{3}\,b_2\right)x_2+b_2x_3-\frac{5}{18\alpha}\left(12b_1^2-4\sqrt{3}\,b_1b_2+13b_2^2\right),
\end{equation*}
where $\alpha\neq 0$, $\psi_0\neq 0$, $b_i$, $i=1,2$~--- arbitrary constants

Now let's move on to constructing exact solutions to ODEs with delay (\ref{9}). Here we have to consider two cases ${\rm tr}\,A+\displaystyle\frac{1}{\lambda\sigma}=0$ and ${\rm tr}\,A+\displaystyle\frac{1}{\lambda\sigma}\neq 0$.

{\bf 1}.$\,$ Let the elements of the main diagonal of the desired matrix $A$ be such that its trace ${\rm tr}\,A$ satisfies the equality
\begin{equation}\label{19}
{\rm tr}\,A+\frac{1}{\lambda\sigma}=0,
\end{equation}   
in this case, we must look for the solution of the matrix equation (\ref{8}) loaded with the condition (\ref{19}).
Taking into account the formula (\ref{11}), the additional condition (\ref{19}) on the trace of the matrix $A$ is equivalent to the equality 
$\lambda=-2/m$. Then the delay ODE (\ref{9}) becomes simpler and takes the form
\begin{equation}\label{20}
\psi'(t)=-\frac{2\alpha}{m}\,\psi^{1+\frac{m}{2}}(t)\psi^{-\frac{m}{2}}(t-\tau),
\end{equation}
where $m\in\{1,2,\ldots ,n\}$,  $m\leq n$, $n\in\mathbb{N}$, $n\geq 2$. The equation (\ref{20}) can be rewritten as
$\left(\psi^{-\frac{m}{2}}(t)\right)'=\alpha\psi^{-\frac{m}{2}}(t-\tau)$ or 
\begin{equation}\label{21}
z'(t)=\alpha z(t-\tau),
\end{equation}
that is, it becomes linear with respect to the new function $z(t)=\psi^{-\frac{m}{2}}(t)$. The linear delay ODE (\ref{21}) has a general solution $z(t)=z_0 e^{\mu t}$, $z_0={\rm const}\neq 0$, where the parameter $\mu$ is from the transcendental equation $\mu e^{\mu\tau}=\alpha$, whose solution is expressed in terms of the special $W$-- Lambert function. Finally, we obtain the exact solution of the equation (\ref{20}) of the following form
\begin{equation*}
\psi(t)=\psi_0\exp\left(-\frac{2\,{\rm LambertW}(\alpha \tau)}{m\tau}\,t\right),
\end{equation*}
where $\psi_0\neq 0$~--- is an arbitrary constant, in particular, $\psi_0=\psi(t)|_{t=0}$~--- is an arbitrary constant. Thus, in this case, taking into account the theorem~\ref{Th1}, the assertion is true.

\begin{state}\label{STh1}
Nonlinear equation with delay 
\begin{equation*}
u_{t}=\nabla\cdot\biggl(u^{-\frac{2}{m}}\nabla u\biggr)+\alpha\bar{u},\quad\bar{u}=u({\bf x},t-\tau ),
\end{equation*}
where $m\in\{1,2,\ldots ,n\}$, $m\leq n$, $n\in\mathbb{N}$, $n\geq 2$ has the exact solution
\begin{equation*}
u({\bf x},t)=\left[l_0\exp\left(-\frac{2\,{\rm LambertW}(\alpha \tau)}{m\tau}\,t\right)\left(\frac{1}{2}(A{\bf x},{\bf x})+({\bf B},{\bf x})+\sigma|{\bf B}|^2\right)\right]^{-\frac{m}{2}},
\end{equation*}
where the numerical symmetric matrix $A$ has the form $(\ref{10})$, and the vector ${\bf B}$ is determined from the linear {\rm SAE} ${\bf B}=2\sigma\,A{ \bf B}$, $l_0\neq 0$~--- arbitrary constant.
\end{state}

{\bf Example 2.}
Let $m=2$, then for $\lambda=-1$ the nonlinear equation with delay has a particular exact solution anisotropic in space variables   of the following form
\begin{equation*}
u_t=\Delta\ln u+\alpha\bar{u}, \quad\bar{u}=u(x,y,z,t-\tau),
\end{equation*}
by state~\ref{STh1} 
\begin{equation*}
u(x,y,z,t)=\left[\psi_0\exp\left(-\frac{{\rm LambertW}(\alpha \tau)}{\tau}\,t\right)W(x,y,z)\right]^{-1},
\end{equation*}
where  
\begin{equation*}
W(x,y,z)=\frac{1}{256\sigma}\left(55\,x^2+39\,y^2+34\,z^2-30\,xy-6\sqrt{30}\,xz-10\sqrt{30}\,yz\right)-
\end{equation*}
\begin{equation*}
-\frac{1}{3}(5\,l_1+\sqrt{30}\,l_2)\,x+l_1y+l_2z+\frac{\sigma}{9}\biggl(34\,l_1^2+10\sqrt{30}\,l_1l_2+39\,l_2^2\biggr),
\end{equation*}
where $\alpha\neq 0$, $\psi_0\neq 0$, $\sigma\neq 0$, $l_1$, $l_2$, $l_3$~--- arbitrary parameters.

{\bf Example 3.}
Let $m=3$, then $\lambda=-2/3$ and the nonlinear  equation with a linear delayed source (sink) in three-dimensional coordinate space of the following form
\begin{equation*}
u_{t}=\nabla\cdot\biggl(u^{-\frac{2}{3}}\nabla u\biggr)+\alpha\bar{u},\quad\bar{u}=u(x,y,z,t-\tau),
\end{equation*}
by state~\ref{STh1} has a particular exact radially symmetric solution
\begin{equation*}
u(x,y,z,t)=\left[\psi_0\exp\left(-\frac{2\,{\rm LambertW}(\alpha \tau)}{3\tau}\,t\right)W_0(x,y,z)\right]^{-\frac{3}{2}}, 
\end{equation*}
where 
\begin{equation*}
W_0(x,y,z)=\frac{1}{4\sigma}\left(x^2+y^2+z^2\right)+l_1x+l_2y+l_3z+\sigma\left(l_1^2+l_2^2+l_3^2\right).
\end{equation*}
where $\alpha\neq 0$, $\psi_0\neq 0$, $\sigma\neq 0$, $l_1$, $l_2$, $l_3$~--- arbitrary parameters.


If the delay parameter $\tau$ is chosen as a fixed positive number, then the general solution of the linear ODE with delay (\ref{21}) can be easily written as a linear combination of sine and cosine. Let $\tau=-\displaystyle\frac{(4k+1)\pi}{2\alpha}$, $\alpha<0$, $k$ -- be any non-negative integer, then the general solution of the equation (\ref{20}) has the form
\begin{equation*}
\psi(t)=\left[C_1\cos(\alpha t)+C_2\sin(\alpha t)\right]^{-\frac{2}{m}},
\end{equation*}
where $C_1$, $C_2$~--- are arbitrary constants such that $C_1C_2\neq 0$.

{\bf 2}.$\,$ Let us start constructing exact solutions to ODEs with delay (\ref{9}) in the general case, assuming that the inequality ${\rm tr}\,A+\displaystyle\frac{1}{\lambda\sigma}\neq 0$ holds, or, taking into account the formula (\ref{11}),
condition $\lambda\neq -2/m$. By a simple change $\psi(t)=\theta^{\lambda}(t)$ we rewrite the equation (\ref{9}) as a relation for the new function $\theta(t)$ 
\begin{equation}\label{22}
\theta'(t)=\frac{a}{\lambda}\,\theta^{\lambda+1}(t)+\alpha\theta(t-\tau),
\end{equation}
where $a\equiv{\rm tr}\,A+\displaystyle\frac{1}{\lambda\sigma}=\displaystyle\frac{1}{\sigma}\left(\displaystyle\frac{m}{2}+\displaystyle\frac{1}{\lambda}\right)\neq 0$~---  constant. In the case of $\lambda=-1$ ($m\neq 2$), the equation (\ref{22}) becomes linear and its general solution has the form
\begin{equation*}
\theta(t)=\theta_0\exp\left(\frac{{\rm LambertW}(\alpha \tau)}{\tau}\,t\right)+\frac{1}{\alpha\sigma}\left(\displaystyle\frac{m}{2}-1\right),
\end{equation*}
where $\theta_0\neq 0$~--- arbitrary constant.

{\bf Example 4.}
Let $\lambda=-1$, $m=3$, then the nonlinear equation with a linear delayed source (sink) in a three-dimensional coordinate space of the following form
\begin{equation*}
u_{t}=\nabla\cdot\biggl(u^{-1}\nabla u\biggr)+\alpha\bar{u},\quad\bar{u}=u(x,y,z,t-\tau),
\end{equation*}
by Theorem~\ref{Th1} has a particular exact radially symmetric solution
\begin{equation*}
u(x,y,z,t)=\frac{\theta_0\exp\left(\displaystyle\frac{{\rm LambertW}(\alpha \tau)}{\tau}\,t\right)+\displaystyle\frac{1}{2\alpha\sigma}}{W_0(x,y,z)}, 
\end{equation*}
where 
\begin{equation*}
W_0(x,y,z)=\frac{1}{4\sigma}\left(x^2+y^2+z^2\right)+l_1x+l_2y+l_3z+\sigma\left(l_1^2+l_2^2+l_3^2\right).
\end{equation*}
where $\alpha\neq 0$, $\theta_0\neq 0$, $\sigma\neq 0$, $l_1$, $l_2$, $l_3$~--- arbitrary parameters.

\section{Conclusion}

In conclusion, we briefly formulate the main results. The article studies the problem of constructing exact solutions to a nonlinear equation of parabolic type with delay, considered as a mathematical model for changing the density of a distributed formation of autonomous robots. A variant of the reduction method is proposed, which makes it possible to separate the search for the dependence of the solution on time and on spatial variables. The dependence on spatial variables is found from a system of algebraic equations, and the dependence on time from a system of ordinary differential equations with delay. A number of examples are given of constructing exact solutions that are explicitly expressed in terms of elementary functions and the Lambert function. As noted in the review~\cite{eb}, most recently partial differential and delayed equations have been used for modelling the movement of formations of interacting robots in swarm robotics, while the main difficulty is that <<analytical solutions of PDEs are available only for a small number of special cases>>. Therefore, the exact solutions of delay PDEs found in the article will be useful in the study of swarm robotics models.


\bigskip

\end{document}
